



Article

## An Approach to Applying the Ability-Motivation-Opportunity Theory to Identify the Driving Factors of Green Employee Behavior in the Hotel Industry

Adina-Raluca Sibian \* and Ana Ispas

Faculty of Economic Sciences and Business Administration, Transilvania University of Braşov, 29, Eroilor Str., 500036 Brasov, Romania; ispasana@unitbv.ro

\* Correspondence: sibian.adinaraluca@yahoo.com; Tel.: +40-0741-642-443

Abstract: In the context of climate change and all other harmful effects of pollution, companies should improve their environmental performances. As part of their strategies, companies should explore the consequences of their environmental practices from the perspective of internal stakeholders and must be more attentive to employee environmental behavior. This study's purpose was to identify the motives that encourage the employees of a green hotel to adopt pro-environmental behaviors. Qualitative research was conducted using semi-structured interviews among three-star hotel employees from Braşov, Romania. This research uses the ability-motivation-opportunity (AMO) theory that could lead to a better understanding of employee green behavior. Data analysis revealed that less than half of the respondents have a proper understanding of the concept of environment and only a third of them are aware of the harmful effects caused by people, both at work and at home. Moreover, most of the respondents know little of the measures implemented by the hotel in order to reduce resource consumption, the extrinsic motivation being dominant. Consequently, the results lead to the recommendation that the hotel managerial team should invest in the employees' training and motivation, creating a corporate environment and a proper organizational culture to develop green behaviors among the employees. The results of this study should be a starting point for those hotels, which face the challenge of involving their employees in the process of greening their activity.

**Keywords:** employee green behavior; green hotel; ability-motivation-opportunity theory; intrinsic motivation; extrinsic motivation

## 1. Introduction

The literature defines employee green behavior (EGB) as a pro-environmental behavior performed by an organization's employees [1–3]. Ramus and Killmer's study [4] has raised interest in the subject of EGB. Researchers' concern regarding the study of employees' behavior at the workplace is explicable as it has been proven that individuals spend 60–70% of their time at work [5]. Moreover, it has been shown that non-residential buildings are responsible for 20% of the carbon emissions at the global level [6]. It has been demonstrated that the employees' involvement is crucial for companies to be able to implement the environmental standards (including ISO 14001) [7–10]. However, there are some EGBs that are not described by official documents [7]. Some examples were identified in Yuriev et al.'s paper [10]: recycling, turning off lights and electric appliances, opting for video conferences instead of traveling, using public transportation, or drinking from reusable cups and bottles. These activities may seem less significant, but in time, they become important for the environmental performances of a company [11–13]. These apparently less significant behaviors, sometimes, go unnoticed by the staff in high hierarchies because of their spontaneous nature [14].

The tourism sector is responsible for approximately 5% of the global carbon emissions, out of which 20% are caused by accommodation services [15] (1% of the global



Citation: Sibian, A.-R.; Ispas, A. An Approach to Applying the Ability-Motivation-Opportunity Theory to Identify the Driving Factors of Green Employee Behavior in the Hotel Industry. Sustainability 2021, 13, 4659. https://doi.org/10.3390/su13094659

Academic Editors: Marc A. Rosen and Ninh Nguyen

Received: 27 February 2021 Accepted: 20 April 2021 Published: 22 April 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/licenses/by/4.0/).

Sustainability **2021**, 13, 4659 2 of 19

emissions [16]). Therefore, the hotel industry should initiate activities that, in time, can contribute to the decrease in its negative impact on the environment [17–19]. Most environmental management practices are implemented in hotel units from developed countries, as demonstrated in previous studies regarding this matter (e.g., in Spain [20–22]; in Greece [23], and in the USA [24]). In leading mature tourist destinations (such as Spain, Italy, France, Turkey etc.), measures to reduce energy consumption or waste part of the day-to-day basic hotel operation [17,21,22,25–35], being part of the hotel culture, mission and vision and are mainly supported and awarded by tour operators [31], local governments and international associations (e.g., the annual Environmental Award of the International Hotel & Restaurant Association—IH&RA [36]).

In the hospitality industry, the employees are the interface between the accommodation facilities and the customers and can be ambassadors of green behavior. Because of that, it is of real interest to study the reasons that may determine them to embrace a green behavior. Moreover, it has been found that EGB in the hotel industry has an important influence on tourists' perception regarding the quality of the environmental practices in the accommodation facilities [37]. That means that, indirectly, the attitude of the tourists regarding the protection of the environment can be influenced by the eco-friendly behavior of the employees [17].

Previous research has focused on mature tourist destinations located in Western countries and on the discovery of the influencing factors on EGB, such as the motivation of the employees [19,38], the perceived organizational support [39], the environmental policies of the company [40], the conscientiousness of the employees, their knowledge and awareness regarding the environmental issues [41,42], the human resources practices [43], and ethical leadership [3,44]. However, few studies have focused on East-European post-communist countries, which are characterized by specific employees' behavior (e.g., [45]).

Romania is an emerging tourist destination in which the lodging industry has developed rapidly in the last three decades [46]. Yet, in Romania, there is a local absence regarding the implementation of green measures to save resources and protect the environment in the hospitality area [47,48]. Moreover, Romanian citizens give little importance to the environment, providing poor performance on the matter of environmental concern and protection [49,50]; generally, there is a lack of citizens' eco-friendly consciousness and knowledge [51]. In addition, there is a tendency related to the managerial organization in which well-structured training programs and effective communication between employees in Romania are not adopted [51,52]. To our knowledge, in Romania, other studies regarding environmental improvements in accommodation units [53–55] have focused only on the perspectives of accommodation managers and tourists, neglecting the value of employee involvement in greening the activities. Consequently, more studies are needed to deepen the knowledge of what can motivate Romanian employees to be actively involved in environmental protection. This research aims to compensate for this lack.

There are plenty of theoretical models used in researching EGB, such as self-determination theory used to analyzes the motivation of the employees [56–60], the theory of planned behavior [58,61–65], social exchange theory [66–68], or ability-motivation-opportunity (AMO) theory [69,70]. Human resource (HR) literature has considered the use of the AMO theory to highlight the connection between HR practices and performance [70]. Moreover, some authors have found it useful to apply AMO theory to improve the systematic perspective in order to enrich the understanding of organizational issues regarding the implementation of HR practices [70]. For this study, the AMO theory was chosen, which emphasizes EGB from the perspective of willingness, consciousness, and habits [10,70] because it appears that only a little research has been conducted to certify this model [70]. A comprehensive exploration of the model should result in a better understanding of the issues that companies face when trying to implement HR [70].

In order to achieve clearer results, each dimension of the AMO theory was analyzed separately. For each component of the AMO theory, an operational objective was phrased, and to each operational objective, a hypothesis was linked.

Sustainability **2021**, 13, 4659 3 of 19

In order to fulfill these objectives, qualitative research was used, as the literature focuses more on using quantitative ones (78%) to the detriment of the qualitative ones (9%) or the mixed ones (12%) [10]. Yuriev et al.'s [10] criticism regarding the lack of differentiation (in 86% of the cases) between employee types when it comes to EGB research called for a focus on non-decision-making employees. With consideration to the above, it was decided to use a semi-structured interview model among the hotel employees from all departments except the administrative ones.

This article is structured as follows: Section 2 provides the literature review of the variables that influence EGB, with the main emphasis on the AMO theory; Section 3 presents the methodology used in the development of this study; Section 4 explains the results from the data collected, structured according to the established objectives; Section 5 includes discussions of the results, conclusion and practical implications of this research. Lastly, Section 6 draws limitations and future research directions.

# 2. Literature Review of the Variables That Influence EGB: A Theoretical Framework for Using AMO Theory

The resource-saving behavior that a person exhibits differs between the private environment (their home) and the workplace because, as an employee, the person does not have access to information regarding the amount of consumption and is not the one who bears the expenses. Factors such as organizational values, managerial support, colleagues' attitudes, and internal culture may influence the pro-environmental behaviors of individuals at work [10]. However, there are usually several employees at work who work in the same position, use the same equipment and consumables, so it is difficult to identify personal consumption, and because of this, it is difficult to determine a level of personal efficiency in saving resources [17]. In addition, people often think that it is the duty of authorities or business managers to take care of environmental issues [19].

"Going green" implies higher efforts besides implementing eco-friendly activities or launching initiatives [28]. Some authors [71] have expressed that there is a crucial need to examine environmental practices and their repercussions from the point of view of internal stakeholders (i.e., employees) because reducing a company's environmental impact depends on the employees' willingness to adopt a green behavior [14,19]. In 2003, Rothenberg [28] mentioned that employees are essential when it comes to integrating the environmental dimension in the fundamental activities of the company. An example was given by Savage [28], referring to the serving staff who knows the level of food wastage better than the hotel managers. Consequently, it is of great importance to create a motivating and supportive environment for the employees. EGB does not contribute only to the improvement of the environmental performances but also to the implementation of sustainable development [72], to the improvement of the efficiency and satisfaction of the employees [73], as well as to decrease the chances of them leaving the company [15].

Although it is well established that employee involvement is particularly important when it comes to "greening" an organization's activity [66–69], the question arises regarding what motivates employees to engage in pro-environmental behaviors [45]. Earlier research has shown that individual and organizational factors combine in a holistic framework that influences the employees' engagement in pro-environmental behaviors [14,17,66,68,74,75]. Many studies have revealed that personal environmental values or beliefs determine EGB in the workplace [12,41]. In addition, individuals' behavior is determined, to a large extent, by the milieu in which they function [45], the "organizational climate", which is defined as employees' shared perceptions concerning organizational policies, practices, and procedures [76]. In fact, "the dimensions that determine the green climate include environmental policy and management orientation (e.g., policy statements, training, information released to employees and supervisor's behavior) as well as specific environmental tasks (e.g., recycling, water resource management and chemical control)" ([17], p. 438).

One of the widely accepted theories to explain the connection between human resources management and employee performance is the ability, motivation, and opportunity (AMO) theory [70]. "Employee performance is considered to be a function of

Sustainability **2021**, 13, 4659 4 of 19

ability/capacity to perform (including different variables such as age, knowledge, level of education and energy level), willingness to perform (including variables such as motivation, job satisfaction, personality, values, and expectations), and opportunity to perform (that included variables such as working conditions, tools, materials, leader behavior, procedures and time)" ([70], p. 1042). There are authors [69] who have focused on the AMO theory, considering the three dimensions (ability, motivation, opportunity) as significant in influencing EGB.

The ability dimension is usually defined by the acronym KSA (knowledge, skills, and abilities) [77]. Employees must be able to recognize and understand environmental issues and try to minimize negative effects [69]. Environmental knowledge is considered to determine the ability to understand and assess the impact of society on ecosystems. Environmental awareness arises from knowing the impact that man has on the environment. Some authors [69] consider that ability is correlated with the level of training of employees. The role of training is to provide employees with the right knowledge, values, and attitudes to achieve the environmental objectives of the organization. Moreover, Gao et al. [5] state that individuals with a higher degree of education frequently tend to show behavior inclined to protect the environment and thus save resources. Training activities have a role both in the qualification of employees and in their motivation [78]. When environmentally responsible behavior becomes a habit, it is much more likely for the individual to take the initiative and repeat this behavior in various forms, thus leading to an increase in the intention to implement environmentally friendly practices in the future.

Motivation deals with an employee's desire to perform, which can be enhanced by extrinsic or intrinsic motivation. Intrinsic refers to something that belongs to the person naturally, integrated, as opposed to the concept of extrinsic, which refers to something that comes from the outside [79]. In general, employees' motivation to work is considered an important factor for the company's success, which is influenced by economic, psychosocial, and organizational factors [78]. Other authors identify attitudes, social norms, and constraints as factors influencing motivation [80]. Having a positive attitude also increases the employees' intention to act, which is a prediction for a certain type of behavior [81], including EGB. The authors see in this intention to act and in the expectations that employees have toward themselves the main variables that influence the behavior of workers regarding energy saving. The idea of the relationship between motivation and attitude is also strengthened by other authors [19], who consider that motivation is the primary element related to employees' attitudes and behavior. Social exchange theory indicates that when an employee perceives certain benefits from the organization, they will feel the need to provide an exchange of service [69]. This principle is also the foundation of motivation.

The motivation to have an environmentally friendly behavior is given not only by personal predisposition and values related to environmental protection but also by the organizational culture of the company, the existence of equipment and technical solutions to save resources, as well as the support provided by the management staff. In other words, it is about the opportunities created at work.

The contribution of the organization in determining employees to save resources and protect the environment is seen through the involvement of managers, leadership, and the effectiveness of communication at the company level [82]. Environmental leaders can strengthen employees' internal motivation by bringing the company's environmental values and goals closer to them [83]. Jerónimo et al. [84] have concluded that employees are prone to engage in activities relevant to environmental protection if their perception of the company's involvement and importance to environmental activities is positive, in other words, if given the opportunity. This aspect is explained by the same authors: if colleagues or superiors show environmentally responsible behavior, individuals are prone to follow this pattern of behavior. Kornilaki and Font [85] have reached similar conclusions, noting that individuals tend to make decisions that fit the social norms of the group in which they work in order to identify with the group. Behavior is shaped to allow integration. Employees may be motivated to behave in an environmentally friendly

Sustainability **2021**, 13, 4659 5 of 19

manner if this behavior is visible in a group (the group of colleagues) that appreciates such manifestations. On the other hand, individuals may exhibit behaviors that are against good morals related to the environment when the group they belong to capitalizes more on the lack of ethics [85].

The organizational culture and internal policies of companies have been identified as exerting influence on employees, their motivation increasing under the social impact of group norms and depending on the degree of integration [6]. Employees need to know the company's mission, vision, and environmental objectives in order to be motivated to engage in tandem with them [78]. Therefore, training, performance management, reward, and active employee involvement are identified as green human resource management practices [69]. Some authors consider that AMO influence in employee's performance is more complicated than expected because it depends not only on the existence of a set of these practices but also on the employee's subjective perceptions of these practices [70].

The current context requires tourism companies to take responsibility and rethink their approach to the processes they carry out. The ultimate goal, in the long term, is to obtain both economic benefits and benefits in terms of business sustainability. Employees are the interface of accommodation units with customers and ambassadors of resource-saving and environmental protection behavior, which is why examining the reasons that would lead them to behave in an environmentally friendly manner is a topic of real interest.

#### 3. Materials and Methods

A qualitative approach was adopted in this research in order to understand and explain the driving factors of green employees' behavior. Therefore, a green hotel from Braşov, Romania, was selected for the current study because of its reputation for environmental protection as being a part of the NeZEH project (Nearly Zero Energy Hotels). There are 55 hotels in Brasov, of which only 2 have expressed their intention to go green through the NeZeh program. Only the employees of the hotel that has already implemented the measures specified in the energy audit were analyzed, given the fact that the other hotel is in the early stages of implementation. From the point of view of concern about environmental issues, the chosen hotel has invested heavily in implementing systems that facilitate the reduction of consumption so that today it can be considered a green hotel. The results were quantified, noting the reduction of the CO<sub>2</sub> emissions of the hotel by 60.4%, electricity consumption by 40%, and heat consumption by 35%, as shown by the data published on the company's website. Among the measures adopted, the following are worth mentioning: building insulation, use of solar panels for hot water production, use of 2-stage power taps in order to reduce water consumption, use of LED lighting, use of a card-based electricity operating system, thus eliminating the risk of unjustified consumption, use of lighting based on motion sensors, use of condensing boilers, and environmentally friendly personal care products.

The measures that have been implemented so far are urgently needed in order to protect the environment and reduce resource consumption, but not enough in the absence of human contribution. Both the hotel staff and the guests who cross its threshold should behave responsibly and support the efforts that have been made so far.

The purpose of this research was to identify the motives that encourage the employees of a green hotel to adopt pro-environmental behaviors. With this aim, exploratory qualitative research was carried out in order to understand employees' knowledge, awareness, and concern regarding environmental issues, as well as their perception regarding the opportunities created by the company to support greening the hotel.

The research method was the interview. In-depth interviewing was chosen as the investigation technique. The used procedure was the semi-structured interview, and the interview guide (see Appendix A) was used as an investigative tool and was developed, taking into consideration the main aspects that define the purpose of this research. The interview guide was designed using the stair-climbing technique [86], the questions being asked in a logical sequence that exhausts all the aspects closely related to the investigated

Sustainability **2021**, 13, 4659 6 of 19

subject [87]. The questions were designed based on the determinants highlighted by the literature review (see the previous chapter) but adapted to fit the context. They were grouped into three parts, each part corresponding to one of the three operational objectives and trying to test the corresponding hypothesis (see Table 1). The hypotheses were formulated based on theoretical background and deduction.

AMO THEORY COMPONENTS	OPERATIONAL OBJECTIVES	HYPOTHESES  Hypotheses 1 (H1). Hotel employees are prone to environmentally friendly behavior.	
A—Ability	O1. Identifying the personal tendency of the hotel employees to adopt an environmentally friendly behavior.		
M—Motivation	O2. Identifying the main motivations that can determine the hotel's employees to adopt a green behavior.	<b>Hypotheses 2 (H2).</b> <i>Intrinsic motivation</i> has the strongest impact on employee involvement in environmental protection.	
O—Opportunity	O3. Identifying employees' perception regarding the company's involvement in greening its activity.	<b>Hypotheses 3 (H3).</b> Hotel employees are aware of the company's involvement in greening its activity.	

Table 1. Operational objectives and hypotheses.

The interviews were conducted by telephone, with two exceptions from which written answers were received. On average, the interviews lasted for about an hour. Respondents were introduced to the topic of discussion and how the interview is to be conducted. Respondents were informed that they were expected to respond honestly, based on their own experiences, and that there were no right or wrong answers in order to create a state of detachment. The discussions occurred in Romanian so as to allow a clearer understanding of the content, and they were recorded and subsequently transcribed verbatim to make it easier to analyze, for which the interviewees were asked for their consent. Nvivo 12 software was used to facilitate the transcriptions and the coding process. Respondents were informed that the information collected is confidential and will be used only for scientific purposes.

The sample base was made up of all hotel employees from March–May 2020. The quota method was chosen for the sampling process because it has several advantages in qualitative methods [88]. The respondents had to be employees who do not have a managerial position and, in terms of the activities they carry out, have access to the hotel's resources, contribute to the resulting level of consumption, and are compatible with the general purpose and objectives of the research. Moreover, the respondents were chosen from all departments covering all the occupational categories of the hotel: reception, housekeeping, kitchen, service, technical, and accounting. Out of all the employed persons (36 employees), 22 were selected (61.11%). Those that correspond to the selection criteria. Out of the 22 people contacted, only one of them did not show his willingness to be part of the research subjects, thus remaining 21 informants. The structure of the sample of respondents is presented in Table 2.

The information obtained was analyzed based on the content analysis method, using both inductive and deductive approaches. The intention was to observe the majority trend for each answer. The answers were also analyzed in terms of the characteristics of the respondents by age categories, education, sex, marital status, parental status, and the period that has elapsed since the employment in the hotel unit. The analysis involved identifying keywords, repetitive opinions, similarities, and differences between categories of respondents in order to define trends. The aim was to determine whether or not employees have so far behaved attentively toward the resources of the hotel unit, the causes of the presence or absence of this type of behavior (existence or lack of concern for environmental issues, motivation, and the work environment to support such activities) and finally, how it can positively influence the evolution of environmentally friendly behavior.

Sustainability **2021**, 13, 4659 7 of 19

**Table 2.** Profile of informants.

No.	Age Group	Sex	Marital Status	Parental Status	Education	Department	Seniority in the Hotel
R1	31–40	Male	Unmarried	Not a parent	Bachelor's degree	Kitchen	1 year
R2	18–30	Female	Unmarried	Not a parent	Bachelor's degree	Front desk	$1\frac{1}{2}$ year
R3	18–30	Female	Unmarried	Not a parent	High School	Front desk	½ year
R4	18–30	Female	Unmarried	Not a parent	High School	Front desk	2 years
R5	18–30	Male	Unmarried	Not a parent	Bachelor's degree	Front desk	3 months
R6	18–30	Female	Married	Parent	High School	Kitchen	3 years
R7	41–50	Female	Married	Parent	Vocational School	Kitchen	2 years
R8	41–50	Male	Married	Parent	High School	Kitchen	4 years
R9	31–40	Male	Unmarried	Not a parent	Gymnasium	Kitchen	14 years
R10	41–50	Female	Married	Not a parent	Gymnasium	Kitchen	$\frac{1}{2}$ year
R11	41–50	Male	Married	Parent	High School	Service	2 years
R12	41–50	Female	Unmarried	Parent	High School	Service	$2\frac{1}{2}$ years
R13	>50	Female	Married	Parent	High School	Service	$5\frac{1}{2}$ years
R14	>50	Female	Unmarried	Parent	High School	Service	½ year
R15	18–30	Male	Unmarried	Not a parent	High School	Service	$\frac{1}{2}$ year
R16	31–40	Female	Married	Parent	High School	Service	$1\frac{1}{2}$ year
R17	31–40	Female	Unmarried	Not a parent	Bachelor's degree	Accounting	2 years
R18	31–40	Female	Unmarried	Not a parent	Gymnasium	Housekeeping	13 years
R19	18–30	Female	Unmarried	Not a parent	High School	Housekeeping	2 years
R20	41–50	Female	Unmarried	Parent	Gymnasium	Housekeeping	$\frac{1}{2}$ year
R21	>50	Male	Married	Parent	Bachelor's degree	Technical	2 years

Source: the authors' research.

### 4. Results

The centralization of the answers obtained from the employees of the analyzed hotel was structured according to the established objectives. The structure used, on the basis of which the results of the research will be stated below, is shown in Table 2.

4.1. Ability. Identifying General Skills of Employees Regarding Saving Resources and Protecting the Environment

The first operational objective (O1) was to identify the extent to which hotel employees have the necessary knowledge, awareness, and active concern to save natural resources and protect the environment.

Firstly, to familiarize the respondents with the research topic, they were asked to explain what they understand by the term environment. Thus, out of the 21 respondents, 7 used the term nature to define the environment, this being the answer with the highest frequency. Five of the respondents explained the concept of the environment as the space in which they spend their time through expressions such as: "what surrounds us" (R2), "the environment in which we live" (R4), "the place where we live and stay day by day" (R13), "what is around me" (R18). A total of 3 respondents referred to the people they interact with: "people around" (R4), "family, neighbors" (R18), "the world around us" (R20).

Sustainability **2021**, 13, 4659 8 of 19

## 4.1.1. Identifying the Awareness of Employees about the Issues That Negatively Affect the Environment

Regarding the identification of aspects that adversely affect the environment, respondents were encouraged to list as many elements as possible. The evaluation criterion used in this case was the number of examples that the respondents were able to provide. The average number of elements listed by the interviewees was 2.52, which indicates a low level of awareness of environmental issues in the context in which a total of 12 potential answers were identified, including: garbage placed in inappropriate places, pollution caused by cars, deforestation, water contamination, etc.

Awareness of the impact of the hotel on the environment has been closely monitored in the context in which the hotel wants to position itself as a green hotel and tries to eliminate or at least reduce the harmful effects of its activity on the environment as much as possible.

Contrary to the first hypothesis (H1), which considers that employees are aware of the negative implications of the hotel's activity on the environment, it turned out that 14 of the 21 respondents do not consider that the hotel affects the environment in any way. Moreover, 6 of them consider the hotel's activity as beneficial to the environment, on the one hand, because the hotel has already implemented certain environmental protection measures that have been identified by the employees (selective waste collection, use of ecological cleanliness products, reduction of resource consumption), and on the other hand, for various other reasons such as: the hotel is a place where people come to discuss environmental issues (R5), the hotel, by its positioning, brings people into nature (R8), the money of tourists who cross its threshold can be used as a form of contribution to the well-being of nature (R18), and the hotel, through its activity, has a good influence on tourists who can take over certain good practices (R21).

Of the seven respondents who consider that the hotel activity affects the environment, three consider that the impact of the hotel exists, but not at a high level. Four respondents supported the statement that the hotel affects the environment, motivating this statement by the fact that the hotel is a large consumer of resources (energy, gas, food), it generates garbage, it improperly stores waste, and due sewage discharges.

### 4.1.2. Identifying Employees' Concern for Environmental Protection

In order to identify the hotel employees' concerns about environmental issues, respondents were asked to confirm or deny their interest in resource-saving and environmental protection issues and to exemplify through some of the resource-saving measures they have adopted at home. Fourteen of the 21 respondents mentioned that they were interested in the problems faced by the environment; two respondents said they were moderately interested, and four respondents said they were not interested in such issues. Respondents who stated that they are interested in the environment are part of all age categories, sex, and level of education. Respondents who stated that they are not interested in environmental issues have one common feature: that they are not married.

It was verified whether the 14 respondents who stated that they were concerned about environmental issues had taken practical measures at home to reduce resource consumption and the negative impact on the environment. Two of the 14 did not specify any such measures. The answers most frequently referred to the care to maintain the lowest possible consumption of utilities: electricity, gas, and water. Eleven of the 21 respondents stated that they take this into consideration. Some of them confessed that they pay attention to the consumption of resources because this implies lower costs, so the reduction of expenses, as the respondents are in the category of those with low incomes. The low consumption of resources was also justified by a low need at the household level. Out of the four respondents who expressed their lack of interest in environmental issues, two of them maintained their position, stating that there is no measure they take at home in order to reduce the negative impact on the environment. The other two specified that they either sort the garbage or pay attention to resource consumption.

Sustainability **2021**, 13, 4659 9 of 19

Twelve employees of the hotel were considered to be concerned, to some extent, with the reduction of resource consumption and the protection of the environment because they met both the condition of declaring themselves interested in environmental protection and the condition of taking concrete measures in this sense. Percentage-wise, they represent  $\approx 57\%$ , so more than half of the respondents.

# 4.2. Motivation. Identifying the Main Motivations of Hotel Employees to Save Resources and Protect the Environment

The literature attributes positive values to intrinsic rather than extrinsic motivation [82], which led to formulating the second hypothesis (H2) that hotel employees are prone to get involved in saving resources and protecting the environment because of the impulses given by intrinsic motivation.

## 4.2.1. Identifying Intrinsic Motivation

Intrinsic motivation appears to be defined as self-satisfaction, unrelated to external benefits. This type of motivation is associated with positive experiences and personal development through the exercise and expansion of individual abilities [89]. From the perspective of employers, intrinsic motivation could be more advantageous, both in terms of costs and due to the fact that it has been shown to provide better results [90].

During the conversations with the 21 respondents, five forms of manifestation of intrinsic motivation were identified. Of these, job satisfaction has the highest frequency of occurrence, being mentioned by eight respondents, meaning about 38% of respondents. The next most frequent response is personal satisfaction, which refers to the fact that people do not necessarily seek only the appreciation of others but also the fact that they can rejoice in themselves for the positive contribution they have made to saving resources and environmental protection. Four respondents (R6, R7, R10, and R20) have specified this aspect related to their intrinsic motivation. All four respondents are female and have at most a high school diploma.

Personal development was mentioned by two of the respondents (R6 and R8). The R6 respondent stated that during his time at work, he became more concerned about the impact he has on the environment. The R8 respondent, on the other hand, sees his personal evolution manifested by the opportunity he had to learn from the others, an aspect that he would improve in the future during the activity carried out at the hotel. Moreover, R8 is the only one who has mentioned the sense of usefulness as a form of intrinsic motivation. R8 also stands out by the fact that he is the one who has most often expressed himself in favor of intrinsic motivation (3 times). In his words: "I like everything that comes out of my hand to be ok; (then) I am happy." R8 is male, between 41 and 50 years old, is married, has children, and is one of the long-term employees of the hotel (>2 years of employment).

In addition to the R8 respondent, there were 3 other respondents who mentioned twice the contribution of intrinsic motivation for saving resources and protecting the environment. These are R6, R7, and R20, all female. It is interesting to note that both the R8 respondent and the R6 and R7 respondents work in the kitchen. Their intrinsic motivation can be explained by the fact that the results of their work, the dishes, are tangible elements easily evaluated by appearance, taste, smell, so their intrinsic satisfaction is more likely to appear, unlike other colleagues whose work does not result in tangible elements.

## 4.2.2. Identifying Extrinsic Motivation

Ryan and Deci [89] state that extrinsic motivation occurs when people undertake certain activities in order to benefit from contexts separate from themselves. If, in the case of intrinsic motivation, people do certain things for the simple fact that they bring them pleasure and they feel good or accomplished doing them, in the case of extrinsic motivation, people are always looking to gain something from an outside source for performing certain actions.

One finding that was made refers to the fact that extrinsic motivation was dominant in the case of the respondents from the analyzed hotel, which refutes the hypothesis formulated (H2).

Sustainability **2021**, 13, 4659 10 of 19

The appreciation from colleagues, superiors, or clients was the answer with the highest frequency, being mentioned by 9 of the 21 respondents, which means about 43% of all respondents. No common characteristics (age, sex, education, marital status, or parental status) could be identified between these respondents. Other responses were related to the implementation of measures to quantify the results of the effort made by the employees in order to save resources in the hotel (energy, gas, water, consumables). The results of saving resources in the hotel should be communicated to the staff so that the employees realize that their effort was worthwhile and appreciated (measure proposed by 5 respondents,  $\approx$ 24%). There were also 5 respondents who stated that they would appreciate the granting of financial benefits (salary, bonuses) or non-financial as a reward for the effort to save resources ( $\approx$ 24%) and encourage employees to participate in and initiate new ideas for ecological practices.

4.3. Opportunity. Identifying Employees' Perceptions of the Involvement of Hotel Management in Saving Resources and Protecting the Environment

The third operational objective (O3) aimed to identify the employees' perceptions regarding the opportunities created in the hotel in order to develop environmentally friendly behaviors. Given that this is a green hotel, the starting hypothesis was that employees are aware of the concern of the hotel management for saving resources and protecting the environment (H3).

4.3.1. Identifying Employees' Perceptions of Technical Systems, Equipment, and Procedures Implemented in the Hotel in Order to Save Resources and Protect the Environment

To determine whether hotel employees are aware of the environmentally friendly technical and procedural measures implemented over time in the hotel, they were asked: Do you think that the hotel is involved in any way in saving resources and protecting the environment? If so, what measures have you seen implemented?

In this case, the evaluation was made by comparing the answers of the interviewees with the reality of the hotel. Regarding the real situation within the hotel, 13 technical and procedural measures were identified: use of solar panels, building insulation, use of 2-stage faucets, use of LED-based lighting, use of a card-based electricity operating system, lighting based on motion sensors, use of condensing boilers, provision of environmentally friendly personal care products, use of environmentally friendly cleaning products, selective waste collection, collection of used oil, and establishment of working procedures to help reduce resource consumption in various forms in all hotel departments.

Respondents were asked to indicate whether they considered that the hotel had taken measures to reduce resource consumption and protect the environment. Of the 21 employees, four were those who stated that they did not notice the implementation of such measures, which means  $\approx$ 19% of all respondents. The 4 respondents have been working at the hotel for a maximum of 2 years but do not have other common characteristics (R1, R4, R14, and R16).

The 16 respondents who answered affirmatively were asked to list all the measures they reported to be implemented by the hotel in order to reduce resource consumption and protect the environment. The employees identified 10 of the 13 measures mentioned above. The 3 measures that were not specified by any of the respondents were: building insulation, use of condensing boilers, and sensor-based lighting. It is explicable that the first two measures are not known by the respondents, as they are not visible and are measures that require certain technical knowledge, but the lighting based on sensors is an obvious measure. This means either that employees are not paying enough attention to the elements of the environment in which they work or that they are not aware that sensor-based lighting is an important measure for saving electricity in the hotel.

The most frequently occurring response was the selective collection of the waste. Eleven respondents identified it as a measure implemented by the hotel. This response has been followed, in terms of frequency, by procedures that favor saving resources in various forms: saving electricity, gas, water, saving paper, food or cleaning products. Nine respondents specified these procedures.

Sustainability **2021**, 13, 4659 11 of 19

There were two respondents who stood out by identifying a larger number of implemented measures. These are respondents R17 and R19, who both listed 6 methods implemented by the hotel. Others compared to the selective collection and saving itself. Both respondents are female and have been working at the hotel for about 2 years but do not have other similar demographic characteristics.

On average, only 2 resource-saving and environmental protection measures were mentioned out of 13 potential responses. An observation that could be deduced from the conversations with the respondents is that a large number of the employees perceive the measures to reduce the consumption of resources as individual and unrelated measures, not being aware of the bigger picture. The hotel's goal to become a green hotel. According to the answers received, the respondents suggest that the hotel is involved "very little" in saving resources and protecting the environment, that they were presented with measures to reduce resource consumption "only once", that it was done "just a little". It can thus be concluded that the third hypothesis (H3) has not been confirmed.

4.3.2. Identifying Employees' Perceptions of the Involvement of the Management Team in Forming an Organizational Culture of Saving Resources and Protecting the Environment

The next question the interviewees were asked was: Have you ever been trained in environmental issues and how to reduce resource consumption at the hotel? The purpose of this question was to find out if the hotel employees were given the opportunity to benefit from the necessary training to cope with the environmental issues.

"Never, but never, not only here, but at any company I have worked for. I don't think this is a priority in Romania." (R2) is one of the answers received. Fifteen of the 21 respondents have stated that they had not participated in any training on the actions they need to take so that resource consumption is as low as possible. As a percentage, the 15 respondents represent  $\approx$ 71% of the total. Respondents R2 and R3 expressed regret over the lack of training in resource saving and environmental protection. A total of 5 respondents specified that they took part in training sessions. Respondent R5 is one of them, a new hotel employee (<1 year after employment), which may lead to infer that the training procedure for new employees has recently been adapted to form specific skills for saving resources among the employees. This explains why the former employees have not had this experience. Respondents 18 and 19 (R18, R19) also stated that they had taken part in training done at the time of implementation of ecological detergents in order to know how to use them. These two respondents work in the housekeeping department. The fifth respondent (R9), who stated he had been trained in terms of environmentally friendly behavior at work, also confessed that he had been found to be a heavy consumer of water, and he thinks he might have received some specific instructions because of that.

What can be deduced from this information is that the training did not take place in an organized way, involving all employees, with a well-defined purpose of conveying to them that their duty also implies caring about the thoughtful consumption of resources, but rather to remedy certain specific situations in which a waste of resources had been observed.

As stated in the responses to the question: Has any of the people responsible for the hotel management ever told you that the waste of natural resources should be avoided?, the employees had been required to "extinguish, slow down, stop, close", use the natural resources (water, energy, gas) more sparingly, in a constant effort to reduce resource consumption.

Eighteen of the 21 respondents answered affirmatively to this question, most of them alluding to the observations made by the management team of the hotel unit when they noticed an unjustified consumption of resources. The 18 respondents represent  $\approx 86\%$  of the total employees interviewed. On the other hand, there are two respondents who stated that they were not told that they should be careful about resource consumption.

According to the answers received,  $\approx$ 42% of those interviewed identified the employer's requirement to use the resources provided economically as one of the measures implemented to reduce consumption and protect the environment. Analyzing the responses more deeply, it was observed that some of the interviewed employees saw the measure of minimizing the consumption of resources more from the perspective of reducing ex-

Sustainability **2021**, 13, 4659 12 of 19

penses, neglecting the aspect related to hotel ethics, that of protecting the environment and becoming a green hotel. A conclusive example in this regard is provided by respondent R1. He admitted that X (from the management of the unit) once drew his attention to the unjustified consumption in a certain situation, but later stated: "I have never talked about the environment and resources with X".

Jerónimo et al. [84] have noted that employees become prone to engage in environmental protection activities when they have a positive perception of the company's concern in reducing the negative impact on the environment. Based on this idea, the purpose of the questions was to analyze whether hotel employees perceive the work environment as being likely to encourage environmentally concerned behavior in general and resource consumption in particular. In the case of both questions, the answers were affirmative for over three-quarters of the respondents, which indicates that there is a positive influence on the resource-saving behavior exerted through the organizational culture of the company.

## 5. Discussion, Conclusion, and Implications

This study, compared to others in the field, brings forward the perspective of employees working in the tourism sector in a post-communist developing country. The AMO theory was chosen to guide the investigation of the reasons that determine the employees of a hotel unit to behave responsibly in relation to the environment and to be actively involved in the process of reducing resource consumption. Using the context of a green hotel from Romania, the findings revealed that employees have a reduced ability to behave responsibly in relation to the environment. This fact is determined by a lack of knowledge about the concept of the environment, a lack of awareness of the harmful effects caused by the hotel industry, and a failure to take individual responsibility for protecting the environment.

Although the hotel management has created opportunities to reduce resource consumption (by purchasing special equipment and implementing eco-friendly technologies and work procedures), it has failed to develop a green organizational culture. Employees, for the most part, are aware of a few of the implemented measures and do not have a positive perception of the company's concern in reducing the negative impact on the environment. The results of this study show that strategic initiatives cannot be implemented without the active support and participation of individual employees [91]. Consequently, the employees' perspective is essential and can provide practical implications for hotel managers to develop effective tools to determine their employees to get involved in environmental practices.

The first part of the research focused on the "baggage" of knowledge and skills (the ability dimension) that hotel employees have about the environment, starting from the premise that both prior knowledge [92] and concern [93] of individuals on the matter of environmental issues will contribute to better environmental behavior. Identifying the extent to which hotel employees have the necessary knowledge, awareness, and active concern to save natural resources and protect the environment has revealed a number of issues. From the point of view of knowledge, according to the obtained results, less than half of the respondents qualify as people who have a correct understanding of the concept of environment. Regarding the awareness of environmental issues, only six of the respondents, which means  $\approx 29\%$  of the total, proved that they are aware of the harmful effects caused both at work and outside of it. These results lead to the suggestion that the hotel management involves the employees in training sessions [94,95] in order to enrich their eco-friendly knowledge. The employees should be aware of and understand the negative impact of economic activities on the environment in general and of hotel activity in particular.

This research revealed the tendency of the respondents to emphasize what everyone, anyone, people, others, us, Romanians must do, believe or apply. The lasting impression was that the respondents were reluctant to assume individual responsibility. The focus does not seem to be on what I should do as an individual but on mass behavior, in which a sense of responsibility is often "discreet" in its manifestation. Employees should be

Sustainability **2021**, 13, 4659 13 of 19

aware of the hotel vision regarding becoming a green company and perceive it as part of their work responsibility. Together with employees, managers can conceive new work procedures that do not affect the quality of services but contribute to the reduction of resource consumption. As for the future, the hotel should hire new employees that have an ecological behavior [59], as well as considering the organization's sustainability strategies in the recruitment process [3]. For example, during interviews, the ethical values of the applicants may be tested by asking environment-related questions [3].

Further, the study aimed to explore the motivation dimension and its contribution regarding EGB. Both intrinsic and extrinsic motivation are seen as predictors of pro-environmental behavior [96,97]. Intrinsic motivation appears to be defined as self-satisfaction, unrelated to external benefits. This type of motivation is associated with positive experiences and personal development through the exercise and expansion of individual abilities [89]. Among the forms of manifestation of the intrinsic motivation mentioned by the respondents, the following stood out: job satisfaction, personal satisfaction, sense of usefulness, and personal development. These results confirm previous studies in the field [98]. In addition, it was observed that intrinsic motivation has a stronger effect on employees whose results of work are tangible elements, easy to assess by appearance, taste, smell (workers in the kitchen).

Extrinsic motivation was dominant in the responses provided by the green hotel employees. Based on their answers, 13 forms of manifestation of extrinsic motivation were identified. Of these, recognition from the hotel management (in the form of direct feedback) was the response with the highest frequency of occurrence. This result confirms that environmental-based feedback from supervisors can boost employees' motivation to engage in environmental responsibilities [69,75], as well as the fact that, if employees feel that their efforts are noticed by their superiors, they will behave in a more sustainable way [99,100]. Moreover, this result is consistent with ability-motivation-opportunity framework-related arguments from previously published studies [101,102].

This research also highlighted that financial benefits (in the form of salary or bonuses) can be extrinsic motivators (ranked second in frequency of occurrence) that influence employee involvement in resource-saving practices. Moreover, this answer confirms the results of other research [98], namely that some of the hotel employees expected some form of reward because of the savings that they perceived the company has made. Other authors have shown that employees commonly tend to behave in a way they think would make them be appreciated and/or rewarded by their superiors [103]. Therefore, having an incentive program to recognize outstanding environmental performance by employees may be an effective tool to encourage better participation in resource saving [98].

The implementation of measures to quantify the results of the effort made by employees to save hotel resources (energy, gas, water, consumables) was another type of extrinsic motivation mentioned by the interviewed employees. This confirms the results obtained by other authors, who have stated that employees can improve their knowledge of the environment [104,105], understand how their behaviors affect outcomes [64], and meet their environmental responsibility [69] if their involvement in environmental activities can be quantified.

Encouraging employees to get involved in environmental activities as well as encouraging them to bring creative input was another example of extrinsic motivation, supporting the conclusions obtained by other researchers [76,106]. It is less frequently mentioned that if employees perceive that their involvement (through decisions and/or suggestions) regarding environmental matters is taken into consideration, they might be more willing to engage in pro-bono environmental activities, as resulted from the study by Pinzone et al. [105]. Hotel managers need to pay more attention to the evaluation of the employees' environmental performance in order to be able to recognize, encourage and congratulate their efforts. In addition, it can be considered that there should be opportunities for employees to propose saving solutions. These kinds of opportunities should be: suggestion boxes, green groups, and brainstorming.

Sustainability **2021**, 13, 4659 14 of 19

The last part of the research aimed to evaluate the dimension of opportunities created by the hotel, such as the implementation of technologies and working procedures to save resources and protect the environment. Although the hotel management has been involved in reducing resource consumption by purchasing special equipment and implementing eco-friendly technologies and work procedures, employees, for the most part, are aware of a few of these measures. The lack of information from the hotel cannot stimulate the creative contribution of the staff. Without knowing the hotel's vision in terms of protecting the environment, employees risk treating the problem of resource consumption with indifference, resulting in behavior that continues the high consumption of resources, but also preventing the hotel from obtaining the status of a green hotel in the true meaning of the term.

What could be deduced from the answers provided by the employees of the analyzed hotel was that the hotel managers did not carry out an organized training session among all employees, with the well-defined purpose of conveying to them the concern for saving resources, but rather to remedy certain specific situations in which a waste of resources had been observed. In this respect, training for environmental activities is necessary in order to transmit the knowledge and skills needed to improve the employees' environmental performance [69]. This type of training will lead, over time, to encouraging employees to participate in environmental activities [107], enhancing the ability to recognize environmental issues [75], and to understand and minimize negative environmental impacts [108,109].

Evaluating the perception regarding the involvement of the management team in forming a "green" organizational culture highlighted the fact that employees perceive the organizational culture of the company as inclined to reduce resource consumption in order to increase economic efficiency, but not as part of company involvement in protecting the environment. It was observed that some of the interviewed employees saw the measure of minimizing consumption more from the perspective of reducing expenses, neglecting the aspect related to the ethics of the hotel. It can even be stated that, although the hotel aims to become a green hotel, its management has failed to develop a green psychological climate [3]. Jerónimo et al. [84] have noted that employees become prone to engage in environmental protection activities when they have a positive perception of the company's concern in reducing the negative impact on the environment. As such, it is recommended that the hotel management team create a green climate and make employees aware of the company's vision regarding the environmental performance they want to achieve, as employees tend to follow the directions given by their superiors.

#### 6. Limitations and Future Research Directions

This study comes to fill the gaps reported by other researchers [10,62,69] to perform qualitative studies focused on other organizational cultures and countries in order to get rich empirical findings related to the employees' motivations to engage in saving resources within tourism accommodation units. The study has its own limitations, which can be taken into consideration for future research. First, the performance of qualitative research limits the generalization of the obtained results. Second, the research was conducted only among employees of a single company from a single country, Romania. The extrapolation of the results is limited because the research addressed a small number of informants. More study cases, more persons involved, from more hotels, and from different countries, should increase the level of the paper and could help support the conclusions of the paper in a more solid manner. Last, all types of resources were considered, which could hinder the abstracting ability of some employees, who are accustomed to referring only to the type of resources frequently used in the department in which they work.

In order to complement the results obtained, future studies should analyze the vision of the management team on the path that the hotel must follow in order to keep pace with technical progress and environmental management practices that will allow obtaining more and more results, better in terms of saving resources and protecting the environment. Apart from the managerial perspective, future research should also focus on the perspective of tourists on the possibility of their involvement in hotel efforts to reduce resource consumption

Sustainability **2021**, 13, 4659 15 of 19

and environmental pollution. It might also be beneficial to analyze the perception of tourists regarding the perspective of a green hotel in order to find out if the steps taken by the hotel are visible to guests and to what extent they bring a competitive advantage to the company. Using other theoretical frameworks, as well as applying quantitative methods on larger samples of respondents from more hotels, even from more countries, should be taken into consideration for future studies, as they may allow a better extrapolation of the results. Finally, further investigations should focus on the analysis of direct or indirect relations between the dimensions of the AMO framework and the departments where the respondents work.

**Author Contributions:** Conceptualization, A.-R.S. and A.I.; methodology, A.-R.S. and A.I.; validation, A.I.; formal analysis, A.-R.S.; investigation, A.-R.S.; resources, A.-R.S. and A.I.; data curation, A.-R.S. and A.I.; writing—original draft preparation, A.-R.S.; writing—review and editing, A.-R.S. and A.I.; visualization, A.-R.S.; supervision, A.I.; project administration, A.-R.S. and A.I.; funding acquisition, A.I. all authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by Transilvania University of Brașov, Romania.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Data sharing is not applicable to this article.

**Acknowledgments:** We would like to thank the editors and anonymous reviewers for their suggestions and comments to improve the quality of the paper.

**Conflicts of Interest:** The authors declare no conflict of interest.

## Appendix A

#### Interview Guide

#### First Operational Objective

- I. What comes first to your mind when thinking about the environment?
- II. In your opinion, what are the environmental problems we are facing?
- III. Do you think that the activity of the hotel influences the environment in any way?
- IV. Are you interested in environmental issues?
- V. Are there any resource saving measures you have adopted at home? If so, which ones?

### **Second Operational Objective**

- I. What makes you stay at your current job?
- II. What do you think is the most important aspect that gives you satisfaction at work?
- III. What would you improve at your current job?
- IV. What would motivate you to save resources within the company you work for?
- V. Do you think that the appreciation/praise/encouragement received from the management staff could make you pay more attention to saving resources?
- VI. Do you think that a higher salary or other financial rewards can make you more concerned about saving resources?

## Third Operational Objective

- I. Do you think that the hotel is involved in any way in saving resources and protecting the environment? If so, what measures have you seen implemented?
- II. At the hotel, have you ever been trained on environmental issues and how to reduce resource consumption?
- III. Have any of the people in charge ever mentioned to you that the waste of natural resources must be avoided?
- IV. Do you think that the management team of the unit behaves carefully toward the consumption of resources? Are there any role models?

Sustainability **2021**, 13, 4659 16 of 19

#### References

1. Ones, D.S.; Dilchert, S. Environmental Sustainability at Work: A Call to Action. Ind. Organ. Psychol. 2012, 5, 444–466. [CrossRef]

- 2. Norton, T.A.; Zacher, H.; Parker, S.L.; Ashkanasy, N.M. Bridging the gap between green behavioral intentions and employee green behavior: The role of green psychological climate. *J. Organ. Behav.* **2017**, *38*, 996–1015. [CrossRef]
- 3. Saleem, M.; Qadeer, F.; Mahmood, F.; Ariza-Montes, A.; Han, H. Ethical Leadership and Employee Green Behavior: A Multilevel Moderated Mediation Analysis. *Sustainability* **2020**, *12*, 3314. [CrossRef]
- 4. Ramus, C.A.; Killmer, A.B.C. Corporate greening through prosocial extrarole behaviours—A conceptual framework for employee motivation. *Bus. Strat. Environ.* **2007**, *16*, 554–570. [CrossRef]
- 5. Gao, L.; Wang, S.; Li, J.; Li, H. Application of the extended theory of planned behavior to understand individual's energy saving behavior in workplaces. *Resour. Conserv. Recycl.* **2017**, 127, 107–113. [CrossRef]
- 6. Staddon, S.C.; Cycil, C.; Goulden, M.; Leygue, C.; Spence, A. Intervening to change behaviour and save energy in the workplace: A systematic review of available evidence. *Energy Res. Soc. Sci.* **2016**, *17*, 30–51. [CrossRef]
- 7. Boiral, O. ISO 9000: Outside the Iron Cage. Organ. Sci. 2003, 14, 720–737. [CrossRef]
- 8. Boiral, O. Corporate greening through ISO 14001: A rational myth? Organ. Sci. 2007, 18, 127–146. [CrossRef]
- 9. Christmann, P.; Taylor, G. Firm self-regulation through international certificable standards: Determinantsof symbolic versus substantive implementation. *J. Int. Bus. Stud.* **2006**, *37*, 863–878. [CrossRef]
- 10. Yuriev, A.; Boiral, O.; Francoeur, V.; Paillé, P. Overcoming the barriers to pro-environmental behaviors in the workplace: A systematic review. *J. Clean. Prod.* **2018**, *182*, 379–394. [CrossRef]
- 11. Boiral, O. Greening the Corporation Through Organizational Citizenship Behaviors. J. Bus. Ethics 2008, 87, 221–236. [CrossRef]
- 12. Lamm, E.; Tosti-Kharas, J.; Williams, E.G. Read this article, but don't print it: Organizational citizenship behavior toward the environment. *Group Organ. Manag.* **2013**, *38*, 163–197. [CrossRef]
- 13. Tsai, C.-C.; Stritch, J.M.; Christensen, R.K. Eco-Helping and Eco-Civic Engagement in the Public Workplace. *Public Perform. Manag. Rev.* **2016**, 40, 1–25. [CrossRef]
- 14. Daily, B.F.; Bishop, J.W.; Govindarajulu, N. A Conceptual Model for Organizational Citizenship Behavior Directed Toward the Environment. *Bus. Soc.* **2009**, *48*, 243–256. [CrossRef]
- 15. Siyambalapitiya, J.; Zhang, X.; Liu, X. Green human resource management: A proposed model in the context of Sri Lanka's tourism industry. *J. Clean. Prod.* **2018**, 201, 542–555. [CrossRef]
- 16. Coles, T.; Dinan, C.; Warren, N. Energy practices among small- and medium-sized tourism enterprises: A case of misdirected effort? *J. Clean. Prod.* **2016**, *111*, 399–408. [CrossRef]
- 17. Chou, C.-J. Hotels' environmental policies and employee personal environmental beliefs: Interactions and outcomes. *Tour. Manag.* **2014**, *40*, 436–446. [CrossRef]
- 18. Rahman, I.; Reynolds, D.; Svaren, S. How "green" are North American hotels? An exploration of low-cost adoption practices. *Int. J. Hosp. Manag.* **2012**, *31*, 720–727. [CrossRef]
- 19. Zhang, J.; Huang, R. Employees' pro-environmental behaviours (PEBs) at international hotel chains (IHCs) in China: The mediating role of environmental concerns (ECs). *J. Hosp. Tour. Manag.* **2019**, *39*, 129–136. [CrossRef]
- 20. Fraj, E.; Matute, J.; Melero, I. Environmental strategies and organizational competitiveness in the hotel industry: The role of learning and innovation as determinants of environmental success. *Tour. Manag.* 2015, 46, 30–42. [CrossRef]
- 21. Molina-Azorín, J.F.; Claver-Cortés, E.; Pereira-Moliner, J.; Tarí, J.J. Environmental practices and firm performance: An empirical analysis in the Spanish hotel industry. *J. Clean. Prod.* **2009**, *17*, 516–524. [CrossRef]
- 22. Orfila-Sintes, F.; Crespí-Cladera, R.; Martínez-Ros, E. Innovation activity in the hotel industry: Evidence from Balearic Islands. *Tour. Manag.* **2005**, *26*, 851–865. [CrossRef]
- 23. Leonidou, L.C.; Leonidou, C.N.; Fotiadis, T.A.; Zeriti, A. Resources and capabilities as drivers of hotel environmental marketing strategy: Implications for competitive advantage and performance. *Tour. Manag.* **2013**, *35*, 94–110. [CrossRef]
- 24. Zhang, J.J.; Joglekar, N.R.; Verma, R. Exploring Resource Efficiency Benchmarks for Environmental Sustainability in Hotels. *Cornell Hosp. Q.* **2012**, *53*, 229–241. [CrossRef]
- 25. Alonso-Almeida, M.-D.-M.; Rocafort, A.; Borrajo, F. Shedding Light on Eco-Innovation in Tourism: A Critical Analysis. *Sustainability* **2016**, *8*, 1262. [CrossRef]
- 26. Barberán, R.; Egea, P.; Gracia-De-Rentería, P.; Salvador, M. Evaluation of water saving measures in hotels: A Spanish case study. *Int. J. Hosp. Manag.* **2013**, *34*, 181–191. [CrossRef]
- 27. Best, M.N.; Thapa, B. Motives, facilitators and constraints of environmental management in the Caribbean accommodations sector. *J. Clean. Prod.* **2013**, *52*, 165–175. [CrossRef]
- 28. Bohdanowicz, P.; Zientara, P.; Novotna, E. International hotel chains and environmental protection: An analysis of Hilton's we care! programme (Europe, 2006–2008). *J. Sustain. Tour.* **2011**, *19*, 797–816. [CrossRef]
- 29. Gatt, K.; Schranz, C. Retrofitting a 3 star hotel as a basis for piloting water minimisation interventions in the hospitality sector. *Int. J. Hosp. Manag.* **2015**, *50*, 115–121. [CrossRef]
- 30. Erdogan, N.; Baris, E. Environmental protection programs and conservation practices of hotels in Ankara, Turkey. *Tour. Manag.* **2007**, *28*, 604–614. [CrossRef]
- 31. Florido, C.; Jacob, M.; Payeras, M. How to Carry out the Transition towards a More Circular Tourist Activity in the Hotel Sector. The Role of Innovation. *Adm. Sci.* **2019**, *9*, 47. [CrossRef]

Sustainability **2021**, 13, 4659 17 of 19

32. Jacob, M.; Florido, C.; Aguiló, E. Research Note: Environmental Innovation as a Competitiveness Factor in the Balearic Islands. *Tour. Econ.* **2010**, *16*, 755–764. [CrossRef]

- 33. Mak, A.H.; Chang, R.C. The driving and restraining forces for environmental strategy adoption in the hotel Industry: A force field analysis approach. *Tour. Manag.* **2019**, 73, 48–60. [CrossRef]
- 34. Martinez-Martinez, A.; Cegarra-Navarro, J.-G.; Garcia-Perez, A.; Wensley, A. Knowledge agents as drivers of environmental sustainability and business performance in the hospitality sector. *Tour. Manag.* **2019**, *70*, 381–389. [CrossRef]
- 35. Pozo-Barajas, R.; Sánchez-Rivas, J. Relationships between Tourism and Hospitality Sector Electricity Consumption in Spanish Provinces (1999–2013). *Sustainability* **2017**, *9*, 480. [CrossRef]
- 36. Kasim, A. Towards a Wider Adoption of Environmental Responsibility in the Hotel Sector. *Int. J. Hosp. Tour. Adm.* **2007**, *8*, 25–49. [CrossRef]
- 37. Kang, K.H.; Stein, L.; Heo, C.Y.; Lee, S. Consumers willingness to pay for green initatives of the hotel industry. *Int. J. Hosp. Manag.* **2012**, *31*, 564–572. [CrossRef]
- 38. Arnold, J.; Randall, R.; Patterson, F.; Silvester, J.; Robertson, I.; Cooper, C. Work Psychology: Understanding Human Behaviour in the Workplace, 6th ed.; Pearson Education Limited: Harlow, UK, 2016.
- 39. Lamm, E.; Tosti-Kharas, J.; King, C.E. Empowering Employee Sustainability: Perceived Organizational Support Toward the Environment. *J. Bus. Ethics* **2015**, *128*, 207–220. [CrossRef]
- 40. Ramus, C.A.; Steger, U. The roles of supervisory support behaviors and environmental policy in employee "eco-initiatives" at leading-edge European companies. *Acad. Manag. J.* **2000**, *43*, 605–626.
- 41. Bissing-Olson, M.J.; Iyer, A.; Fielding, K.S.; Zacher, H. Relationships between daily affect and pro-environmental behavior at work: The moderating role of pro-environmental attitude. *J. Organ. Behav.* **2012**, *34*, 156–175. [CrossRef]
- 42. Safari, A.; Salehzadeh, R.; Panahi, R.; Abolghasemian, S. Multiple pathways linking environmental knowledge and awareness to employees' green behavior. *Corp. Gov.* **2018**, *18*, 81–103. [CrossRef]
- 43. Saeed, B.B.; Afsar, B.; Hafeez, S.; Khan, I.; Tahir, M.; Afridi, M.A. Promoting employee's pro-environmental behavior through green human resource management practices. *Corp. Soc. Responsib. Environ. Manag.* **2019**, *26*, 424–438. [CrossRef]
- 44. Ko, C.; Ma, J.; Bartnik, R.; Haney, M.H.; Kang, M. Ethical Leadership: An Integrative Review and Future Research Agenda. *Ethics Behav.* **2018**, 28, 104–132. [CrossRef]
- 45. Zientara, P.; Zamojska, A. Green organizational climates and employee pro-environmental behaviour in the hotel industry. *J. Sustain. Tour.* **2016**, *26*, 1142–1159. [CrossRef]
- 46. National Institute of Statistics. Romanian Tourism Statistical Abstract. 2020. Available online: http://www.insse.ro/cms/(accessed on 20 February 2021).
- 47. Băltescu, A.C. The green buildings: Sustainable development actions in the accommodation sector. *Ann. Econ. Ser.* **2018**, 3, 130–135.
- 48. Mester, I.; Badulescu, D.; Badulescu, A.; Bac, D.-P.; Simut, R.; Sehleanu, M. Attitudes and practices concerning sustainability. *A research case on Romanian tourism ventures. J. Environ. Prot. Ecol.* **2016**, 17, 687–695.
- 49. Pirani, E.; Secondi, L. Eco-Friendly Attitudes: What European Citizens Say and What They Do. Int. J. Environ. Res. 2011, 5, 67–84.
- 50. Roman, T.; Bostan, I.; Manolică, A.; Mitrica, I. Profile of Green Consumers in Romania in Light of Sustainability Challenges and Opportunities. *Sustainability* **2015**, *7*, 6394–6411. [CrossRef]
- 51. Todericiu, R.; Şerban, A.; Dumitrașcu, O. Particularities of knowledge worker's motivation strategies in Romanian organizations. *Procedia Econ. Financ.* **2013**, *6*, 405–413. [CrossRef]
- 52. Lupu, N.; Tanase, M.O.; Nica, A.-M. Perception About Business Education in the Tourism Domain and the Hospitality Industry. *Amfiteatru Econ. J.* **2014**, *16*, 800–810.
- 53. Neacșu, N.A.; Bălășescu, S. Using the ECO concept for smart hotel promotion; Case study: Hotel Vega. *Ann. Econ. Ser.* **2017**, 1, 27–35.
- 54. Candrea, A.N.; Brătucu, G. The perspectives for environmental management in small and medium accomodation units. The case of Brașov, Romania. *Manag. Mark.* **2012**, *7*, 465–478.
- 55. Ispas, A.; Untaru, E.N.; Candrea, A.N. Environmental management practices within Agroturism Boarding Houses in Romania: A qualitative study among managers. *Sustainability* **2019**, *11*, 3887. [CrossRef]
- 56. Gagné, M.; Deci, E.L. Self-determination theory and work motivation. J. Organ. Behav. 2005, 26, 331–362. [CrossRef]
- 57. Gatling, A.; Kim, J.; Milliman, J. The relationship between workplace spirituality and hospitality supervisors' work attitudes: A self-determination theory perspectives. *Int. J. Contemp. Hosp. Manag.* **2016**, *28*, 471–489. [CrossRef]
- 58. Graves, L.M.; Sarkis, J.; Zhu, Q. How transformational leadership and employee motivation combine to predict employee proenvironmental behaviors in China. *J. Environ. Psychol.* **2013**, *35*, 81–91. [CrossRef]
- 59. Kim, S.-H.; Kim, M.; Han, H.-S.; Holland, S. The determinants of hospitality employees' pro-environmental behaviors: The moderating role of generational differences. *Int. J. Hosp. Manag.* **2016**, *52*, 56–67. [CrossRef]
- 60. Welters, R.; Mitchell, W.; Muysken, J. Self determination theory and employed job search. *J. Econ. Psychol.* **2014**, 44, 34–44. [CrossRef]
- 61. Adams, E.J.; Esliger, D.W.; Taylor, I.M.; Sherar, L.B. Individual, employment and psychosocial factors influencing walking to work: Implications for intervention design. *PLoS ONE* **2017**, *12*, e0171374. [CrossRef]

Sustainability **2021**, 13, 4659 18 of 19

62. Chan, E.S.; Hon, A.H.; Chan, W.; Okumus, F. What drives employees' intentions to implement green practices in hotels? The role of knowledge, awareness, concern and ecological behaviour. *Int. J. Hosp. Manag.* **2014**, *40*, 20–28. [CrossRef]

- 63. Blok, V.; Wesselink, R.; Studynka, O.; Kemp, R. Encouraging sustainability in the workplace: A survey on the pro-environmental behaviour of university employees. *J. Clean. Prod.* **2015**, *106*, 55–67. [CrossRef]
- 64. Boiral, O.; Talbot, D.; Paillé, P. Leading by Example: A Model of Organizational Citizenship Behavior for the Environment. *Bus. Strat. Environ.* **2015**, 24, 532–550. [CrossRef]
- 65. Yuriev, A.; Dahmen, M.; Paillé, P.; Boiral, O.; Guillaumie, L. Pro-environmental behaviors through the lens of the theory of planned behavior: A scoping review. *Resour. Conserv. Recycl.* **2020**, *155*, 104660. [CrossRef]
- 66. Paillé, P.; Raineri, N. Linking perceived corporate environmental policies and employees eco-initiatives: The influence of perceived organizational support and psychological contract breach. *J. Bus. Res.* **2015**, *68*, 2404–2411. [CrossRef]
- 67. Paillé, P.; Mejía-Morelos, J.H.; Marché-Paillé, A.; Chen, C.C.; Chen, Y. Corporate Greening, Exchange Process Among Co-workers, and Ethics of Care: An Empirical Study on the Determinants of Pro-environmental Behaviors at Coworkers-Level. *J. Bus. Ethics* **2016**, *136*, 655–673. [CrossRef]
- 68. Raineri, N.; Mejía-Morelos, J.H.; Francoeur, V.; Paillé, P. Employee eco-initiatives and the workplace social exchange network. *Eur. Manag. J.* **2016**, *34*, 47–58. [CrossRef]
- 69. Pham, N.T.; Tučková, Z.; Chiappetta Jabbour, C.J. Greening the hospitality industry: How do green human resources management practices influence organizational citizenship behavior in hotels? A mixed-methods study. *Tour. Manag.* **2019**, 72, 386–399. [CrossRef]
- 70. Marin-Garcia, J.A.; Tomas, J.M. Deconstructing AMO framework: A systematic review. *Intang. Cap.* **2016**, *12*, 1040–1087. [CrossRef]
- 71. Choi, H.-M.; Kim, W.G.; Kim, Y.J.; Agmapisarn, C. Hotel environmental management initiative (HEMI) scale development. *Int. J. Hosp. Manag.* **2019**, 77, 562–572. [CrossRef]
- 72. Norton, T.A.; Zacher, H.; Ashkanasy, N.M. Pro-environmental organizational culture and climate. In *The Psychology of Green Organizations*; Oxford University Press: New York, NY, USA, 2015; pp. 322–348.
- 73. Govindarajulu, N.; Daily, B.F. Motivating employees for environmental improvement. *Ind. Manag. Data Syst.* **2004**, 104, 364–372. [CrossRef]
- 74. Daily, B.F.; Bishop, J.W.; Massoud, J.A. The role of training and empowerment in environmental performance: A study of the Mexican maquiladora industry. *Int. J. Operat. Prod. Manag.* **2012**, *32*, 631–647. [CrossRef]
- 75. Schneider, B.; White, S.S.; Paul, M.C. Linking service climate and customer perceptions of service quality: Test of a causal model. *J. App. Psychol.* **1998**, *83*, 150–163. [CrossRef]
- 76. Tian, H.; Zhang, J.; Li, J. The relationship between pro-environmental attitude and employee green behavior: The role of motivational states and green work climate perceptions. *Environ. Sci. Pollut. Res.* **2019**, *27*, 7341–7352. [CrossRef]
- 77. Fu, N.; Flood, P.C.; Bosak, J.; Morris, T.; O'Regan, P. Exploring the performance effect of HPWS on professional service supply chain management. *Supply Chain Manag. Int. J.* **2013**, *18*, 292–307. [CrossRef]
- 78. Ozkeser, B. Impact of training on employee motivation in human resources management. *Procedia Comput. Sci.* **2019**, *158*, 802–810. [CrossRef]
- 79. Chiang, C.-F.; Jang, S. An expectancy theory model for hotel employee motivation. *Int. J. Hosp. Manag.* **2008**, 27, 313–322. [CrossRef]
- 80. Goh, E.; Jie, F. To waste or not to waste: Exploring motivational factors of Generation Z hospitality employees towards food wastage in the hospitality industry. *Int. J. Hosp. Manag.* **2019**, *80*, 126–135. [CrossRef]
- 81. Lopes, J.R.N.; Kalid, R.D.A.; Rodríguez, J.L.M.; Filho, S. Ávila A new model for assessing industrial worker behavior regarding energy saving considering the theory of planned behavior, norm activation model and human reliability. *Resour. Conserv. Recycl.* 2019, 145, 268–278. [CrossRef]
- 82. Okumus, F.; Köseoglu, M.A.; Chan, E.; Hon, A.F.; Avci, U. How do hotel employees' environmental attitudes and intentions to implement green practices relate to their ecological behaviour? *J. Hosp. Tour. Manag.* **2019**, *39*, 193–200. [CrossRef]
- 83. Graves, L.M.; Sarkis, J. The role of the employees' leadership perceptions values and motivation in employees' pro-environmental behaviors. *J. Clean. Prod.* **2018**, *196*, 576–587. [CrossRef]
- 84. Jerónimo, H.M.; Henriques, P.L.; De Lacerda, T.C.; Da Silva, F.P.; Vieira, P.R. Going green and sustainable: The influence of green HR practices on the organizational rationale for sustainability. *J. Bus. Res.* **2020**, *112*, 413–421. [CrossRef]
- 85. Kornilaki, M.; Font, X. Normative influences: How socio-cultural and industrial norms influence the adoption of sustainability practices. A grounded theory of Cretan, small tourism firms. *J. Environ. Manag.* **2019**, 230, 183–189. [CrossRef]
- 86. Malhotra, N.K. Marketing Research: An Applied Orientation; Pearson Education International: Cranbury, NJ, USA, 2004.
- 87. Boyce, C.; Neale, P. Conducting in-Depth Interviews: A Guide for Designing and Conducting In-Depth Interviews for Evaluation Input; Pathfinder International Tool Series: Watertown, NY, USA, 2006.
- 88. Golafshani, N. Understanding reliability and validity in qualitative research. *Qual. Rep.* **2003**, *8*, 597–606. Available online: http://nsuworks.nova.edu/tqr/vol8/iss4/6 (accessed on 24 November 2020).
- 89. Ryan, R.M.; Deci, E.L. Intrinsic and Extrinsic Motivations: Classic Definitions and New Directions. *Contemp. Educ. Psychol.* **2000**, 25, 54–67. [CrossRef]

Sustainability **2021**, 13, 4659

90. Kuvaas, B.; Buch, R.; Weibel, A.; Dysvik, A.; Nerstad, C.G. Do intrinsic and extrinsic motivation relate differently to employee outcomes? *J. Econ. Psychol.* **2017**, *61*, 244–258. [CrossRef]

- 91. Wesselink, R.; Blok, V.; Ringersma, J. Pro-environmental behaviour in the workplace and the role of managers and organization. *J. Clean. Prod.* **2017**, *168*, 1679–1687. [CrossRef]
- 92. Umrani, W.A.; Channa, N.A.; Yousaf, A.; Ahmed, U.; Pahi, M.H.; Ramayah, T. Greening the workforce to achieve environmental performance in hotel industry: A serial mediation model. *J. Hosp. Tour. Manag.* **2020**, *44*, 50–60. [CrossRef]
- 93. Han, H.; Yu, J.; Kim, W. Environmental corporate social responsibility and the strategy to boost the airline's image and customer loyalty intentions. *J. Travel Tour. Mark.* **2019**, *36*, 371–383. [CrossRef]
- Law, M.M.S.; Hills, P.; Hau, B.C.H. Engaging employees in sustainable development

  –a case study of environmental education and awareness training in Hong Kong. Bus. Strategy Environ. 2017, 26, 84

  –97. [CrossRef]
- 95. Perron, G.M.; Côté, R.P.; Duffy, J.F. Improving environmental awareness training in business. *J. Clean. Prod.* **2006**, *14*, 551–562. [CrossRef]
- 96. de Groot, J.I.; Steg, L. Relationships between value orientations, self-determined motivational types and pro-environmental behavioural intentions. *J. Environ. Psychol.* **2010**, *30*, 368–378. [CrossRef]
- 97. van der Werff, E.; Steg, L.; Keizer, K. It is a moral issue: The relationship between environmental self-identity, obligation-based intrinsic motivation and pro-environmental behaviour. *Glob. Environ. Chang.* **2013**, 23, 1258–1265. [CrossRef]
- 98. Chan, E.S.; Hawkins, R. Attitude towards EMSs in an international hotel: An exploratory case study. *Int. J. Hosp. Manag.* **2010**, 29, 641–651. [CrossRef]
- 99. Luu, T.T. Building employees' organizational citizenship behavior for the environment: The role of environmentally-specific servant leadership and a moderated mediation mechanism. *Int. J. Contemp. Hosp. Manag.* **2019**, *31*, 406–426. [CrossRef]
- 100. Temminck, E.; Mearns, K.; Fruhen, L. Motivating Employees towards Sustainable Behaviour. *Bus. Strat. Environ.* **2013**, 24, 402–412. [CrossRef]
- 101. Kim, K.Y.; Pathak, S.; Werner, S. When do international human capital enhancing practices benefit the bottom line? An ability, motivation, and opportunity perspective. *J. Int. Bus. Stud.* **2015**, *46*, 784–805. [CrossRef]
- 102. Van Iddekinge, C.H.; Aguinis, H.; Mackey, J.D.; DeOrtentiis, P.S. A Meta-Analysis of the Interactive, Additive, and Relative Effects of Cognitive Ability and Motivation on Performance. *J. Manag.* **2018**, *44*, 249–279. [CrossRef]
- 103. Moore, C.; Mayer, D.M.; Chiang, F.F.; Crossley, C.; Karlesky, M.J.; Birtch, T.A. Leaders matter morally: The role of ethical leadership in shaping employee moral cognition and misconduct. *J. Appl. Psychol.* **2019**, *104*, 123. [CrossRef]
- 104. Guerci, M.; Longoni, A.; Luzzini, D. Translating stakeholder pressures into environmental performance—the mediating role of green HRM practices. *Int. J. Hum. Resour. Manag.* **2015**, *27*, 262–289. [CrossRef]
- 105. Pinzone, M.; Guerci, M.; Lettieri, E.; Redman, T. Progressing in the change journey towards sustainability in healthcare: The role of 'Green' HRM. J. Clean. Prod. 2016, 122, 201–211. [CrossRef]
- 106. Masri, H.A.; Jaaron, A.A. Assessing green human resources management practices in Palestinian manufacturing context: An empirical study. *J. Clean. Prod.* **2017**, *143*, 474–489. [CrossRef]
- 107. Kim, W.G.; McGinley, S.; Choi, H.-M.; Agmapisarn, C. Hotels' environmental leadership and employees' organizational citizenship behavior. *Int. J. Hosp. Manag.* **2020**, *87*, 102375. [CrossRef]
- 108. Vidal-Salazar, M.D.; Cordón-Pozo, E.; Ferrón-Vilchez, V. Human resource management and developing proactive environmental strategies: The influence of environmental training and organizational learning. *Hum. Resour. Manag.* 2012, 51, 905–934. [CrossRef]
- 109. Sourvinou, A.; Filimonau, V. Planning for an environmental management programme in a luxury hotel and its perceived impact on staff: An exploratory case study. *J. Sustain. Tour.* **2017**, *26*, 649–667. [CrossRef]